Lawrence Berkeley National Laboratory

LBL Publications

Title

The economic value of a centralized approach to distributed resource investment and operation

Permalink

https://escholarship.org/uc/item/9ft490dv

Authors

Carvallo, Juan Pablo Zhang, Nan Murphy, Sean P <u>et al.</u>

Publication Date

2020-07-01

DOI

10.1016/j.apenergy.2020.115071

Peer reviewed

eScholarship.org



Electricity Markets & Policy Energy Analysis & Environmental Impacts Division Lawrence Berkeley National Laboratory

The economic value of a centralized approach to distributed resource investment and operation

Juan Pablo Carvallo^{1,*}, Nan Zhang², Sean P. Murphy¹, Benjamin D. Leibowicz², Peter H. Larsen¹

¹ Electricity Markets and Policy Group, Lawrence Berkeley National Laboratory, 1 Cyclotron Road, MS 90R4000, Berkeley, CA 94720, United States
 ² Operations Research and Industrial Engineering, The University of Texas at Austin, ETC 5.128D, 204
 E. Dean Keeton St., Stop C2200, Austin, TX 78712, United States

*Corresponding Author (jpcarvallo@lbl.gov)

July 2020

This is a pre-print copy of an article published in *Applied Energy*: https://doi.org/10.1016/j.apenergy.2020.115071



This work was supported by the Transmission Permitting and Technical Assistance Division of the U.S. Department of Energy's Office of Electricity under Lawrence Berkeley National Laboratory Contract No. DE-AC02-05CH11231.

DISCLAIMER

This document was prepared as an account of work sponsored by the United States Government. While this document is believed to contain correct information, neither the United States Government nor any agency thereof, nor The Regents of the University of California, nor any of their employees, makes any warranty, express or implied, or assumes any legal responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by its trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or The Regents of the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof, or The Regents of the United States Government or any agency thereof.

Ernest Orlando Lawrence Berkeley National Laboratory is an equal opportunity employer.

COPYRIGHT NOTICE

This manuscript has been authored by an author at Lawrence Berkeley National Laboratory under Contract No. DE-AC02-05CH11231 with the U.S. Department of Energy. The U.S. Government retains, and the publisher, by accepting the article for publication, acknowledges, that the U.S. Government retains a non-exclusive, paid-up, irrevocable, worldwide license to publish or reproduce the published form of this manuscript, or allow others to do so, for U.S. Government purposes.

The economic value of a centralized approach to distributed resource investment and operation

Juan Pablo Carvallo^{1,*}, Nan Zhang², Sean P. Murphy¹, Benjamin D. Leibowicz², Peter H. Larsen¹

¹ Electricity Markets and Policy Group, Lawrence Berkeley National Laboratory, 1 Cyclotron Road, MS 90R4000, Berkeley, CA 94720, United States
² Operations Research and Industrial Engineering, The University of Texas at Austin, ETC 5.128D, 204 E. Dean Keeton St., Stop C2200, Austin, TX 78712, United States

* Corresponding Author (jpcarvallo@lbl.gov)

Abstract

Distributed energy resources have been almost exclusively deployed and operated under a decentralized decision-making process. In this paper, we assess the evolution of a power system with centrally planned utility-scale generation, transmission, distribution, and distributed resources. We adapt a capacity expansion model to represent both centralized and decentralized decision-making paradigms under various electricity rate structures. This paper shows that a centralized planning approach could save 7% to 37% of total system costs over a 15-year time horizon using a Western United States utility as a case study. We show that centralized decision-making deploys substantially more utility-scale solar and distributed storage compared to a decentralized decision-making paradigm. We demonstrate how a utility could largely overcome the complications of decentralized distributed resource decision-making by incentivizing regulators to develop electricity rates that more closely reflect time- and location-specific, long-run marginal costs. The results from this analysis yield insights that are useful for long-term utility planning and electric utility rate design.

Keywords: distributed energy resources; decentralized decision-making; electricity rate structure; power system; resource planning; rooftop solar

Abbreviations

Abbreviation	Description
АТВ	Annual Technology Baseline (NREL annual cost study)
BPS	Bulk Power System
CCGT	Combined Cycle Gas Turbine
DER	Distributed Energy Resources
DG	Distributed Generation
GAP	Grid Access and Planning (model)
IRP	Integrated Resource Plan(ning)
LBNL	Lawrence Berkeley National Laboratory
LSE	Load Serving Entity
NEM	Net Energy Metering
NPV	Net Present Value
NREL	National Renewable Energy Laboratory
OPF	Optimal Power Flow
PNNL	Pacific Northwest National Laboratory
PV	Photovoltaic
RoR	Run-of-river
SCCT	Simple Cycle Combustion Turbine
TOU	Time-of-use

1. Introduction

Decisions about the deployment and operation of distributed energy resources (DER) have almost exclusively been made "behind the meter" and based on the cost of equipment, financing and subsidies available, and electricity rates. Utility planners have generally not been involved in determining where and how much distributed generation should be installed. It follows that the decisions to invest in distributed resources have not stemmed from any coordinated planning effort, and the decisions to operate these resources generally do not respond to any coordinated dispatch process. In this paper, we investigate the economic value of coordinating the planning, procurement, and operation of DER.

To date, distribution utilities have grappled with DER adoption by using the feeder hosting capacity to cap the deployment of distributed generation in locations where system reliability may be adversely affected. Reliability concerns related to DER may include voltage and frequency control, reactive power management, less predictable power output, and bi-directional power flows that increase operational complexity [1]. Utilities occasionally need to upgrade or repower distribution systems and circuits to accommodate DER [2]. DER also benefit distribution systems by reducing losses, deferring system capacity expansion, alleviating feeder loading, and, in some cases, improving reliability for end-use customers [3]. Unfortunately, it is difficult to quantify these benefits because DER investment and dispatch are not coordinated with utility-level investment and dispatch. It should be noted that load

serving entities (LSEs) that make bulk power system (BPS) investment and/or operation decisions have started to track these behind-the-meter resources more closely for their short- and long-term planning [4]. Furthermore, there are a small but growing number of utilities that are considering installing and operating owned behind-the-meter DER, which suggests this study could have immediate real-world applications [5].

The lack of control over any DER deployment and the traditional separation of distribution and BPS planning means that LSEs have limited opportunities to manage uncertainty associated with DER. Consequently, LSEs need to forecast adoption and operation of DER in an effort to characterize the net demand they will face in the future. It follows that decentralized additions of DER into the distribution network are most likely misaligned with the least-cost expansion of the BPS. This misalignment is exacerbated in cases where DER adoption forecasts are inaccurate.

DER adoption has increased substantially over the last 15 years driven by technology cost declines, favorable rate structures including net-metering and increased block pricing, and public awareness [6]. The residential photovoltaic (PV) market has grown 44% *annually* since 2005 and about 2.5% of households in the U.S. have installed a PV system [7]. This aggregate figure masks the fact that distributed PV is much more prevalent in certain states like Hawaii, California, and Arizona where 31%, 11%, and 9% of households have PV as of 2018, respectively. PV is the most prevalent residential distributed generation technology, but there could be substantial adoption of batteries as a distributed resource in the near future due to lower costs and increases in efficiency/capacity [8]. DER adoption is expected to continue to grow rapidly [9], which will only increase the inefficiencies associated with the lack of coordination between bulk power and distribution systems.

In this paper, we evaluate an important, yet understudied topic: *the economic value of fully integrating DER into a long-term utility planning process*. To explore this topic, we follow a typical utility planning process by (1) forecasting demand; (2) accounting for existing and future resources (utility-scale and DER); and (3) running a capacity expansion model to determine the least-cost pathway to meet future demand. We use a tool that evaluates least-cost expansion and operation of generation, transmission, distribution, and DER systems: the Grid Access and Planning (GAP) model [10]. The GAP model was originally programmed as a social planner with a centralized decision-making paradigm. We adapt GAP by splitting the customer (i.e. behind-the-meter) and utility decisions to simulate the decentralized decision-making approach that characterizes the industry. The centralized version of the tool provides the counterfactual to assess the differences with a decentralized decision-making framework associated with DER deployment and operation. Therefore, in this paper, the decentralized scenarios are the "business as usual" or *status quo* cases, whereas the centralized scenario is the simulated counterfactual against which the decentralized scenarios are compared.

Before proceeding, it should be noted that at present it might not be politically or technically feasible to centralize the decisions of deployment and operation of distributed resources. However, this analysis makes an important contribution to the literature by quantifying the economic impact of decentralized investment and operation as they already occur in practice—a topic that has not been evaluated before. More specifically, we measure the differences in technology types, sizes, locations, and operational schemes between centralized and decentralized investment and operation of the power

system over its full scope, from distribution to transmission and generation

Based on this assessment, for example, electricity rates could be designed to achieve specific deployment or operation objectives that are a proxy for potentially more efficient decisions made by a centralized operator. Results from this paper have important policymaking implications as they highlight how long-term planning processes could be improved by integrating bulk power and distribution system planning [11]. This study answers the following specific research questions:

- What are the cost differences between decentralized versus centralized investment in and operation of DER?
- What are the technological and operational differences between a system managed with a decentralized decision-making structure versus a centralized one?
- What impacts do rates have on DER adoption and how does decentralized DER adoption compare to the centralized decision-making outcome under different rate structures?
- What changes could be implemented in the utility planning processes to address the challenges of decentralized decision-making?

This manuscript is organized as follows. Section 2 contains a short review of relevant literature. The GAP model, its adaptation to a decentralized framework, and its calibration are explained in Section 3. Section 4 includes a case study and Section 5 describes key results. Section 6 concludes with several findings that are relevant to utility IRP processes. All figures and tables are our own unless noted.

2. Relevant Literature

Increased adoption of DER in the past decade is driving research into its impacts on distribution, transmission, and generation, and how these impacts are mediated by alternative decision-making paradigms. This paper addresses the second topic by comparing the effect that centralized and decentralized decision-making paradigms have on power system expansion with DER. This section then provides an overview of the literature on the impacts of DER adoption on power systems, with a focus on the physical and economic effects of DER adoption on the planning, design, and operation of power systems.

Research on the impacts of DER in power systems has largely been organized around the traditional divisions between the distribution, transmission, and generation components of the power system. Fewer studies have assessed the cross-component impacts, i.e. in the generation-transmission components or across distribution-generation components. We describe this components-related research in Figure 2.1 below and identify the different areas of research within each component and across components. Areas 1, 2, and 3, represent research focused on DER impacts in either distribution (including behind the meter), transmission, or generation components of the power system. Areas 4, 5, and 6, represent papers that address the impacts of DER on the distribution and transmission, transmission and generation, or generation and distribution interfaces. For example, work that studies how DER adoption could displace utility scale generation would be located in area 6. Area 7 represents papers that concurrently address impacts of DER on the three components of the power system.

Studies have generally focused on DER impacts on area 1 (distribution system) rather than area 2, the transmission system [12]. For example, [13] study the optimal decision making of a distribution utility to minimize power purchases given certain levels of PV adoption. There is no capacity expansion or forward looking component in this work. Technical analyzes have explored the impacts of PV adoption in rural feeders [14], on voltage control in urban networks [15], and of very high levels of adoption [16]. These results do not explore impacts beyond power quality issues at the distribution system level only. Area 5, the generation-transmission interface, has also been subject of substantial work especially on the very short term time horizon. For example, [17] analyzes the dynamic impacts of distributed and utility-scale PV penetration on the transmission system. A similar analysis is developed by [12], concluding that distributed systems are more favorable to system stability than utility-scale systems. However, these paper and literature is generally limited to transient analysis of the transmission system, without extending to distribution and without considering production cost and capacity expansion modeling. [18] and [13] are the only studies found that address DER impacts in area 4, the transmission and distribution sector interface (area 4). In [18], an integrated distribution-transmission model-developed by the National Renewable Energy Laboratory (NREL)-identifies the benefits of being able to forecast the impacts of distributed generation at the transmission level. The forecasts lead to improved reliability measurements and reduced generation costs. [13] does not estimate impacts of DER on the transmission system, but does consider transmission constraints on a model of a distribution company balancing DER dispatch and procurement. There are no known studies that compare the concurrent impacts of DER deployment in operation and investment on all three levels of the power system (area 7 in Figure 2.1) in a single modeling framework. This paper addresses this knowledge gap.

The diversity of papers exploring a high DER penetration future reflects the wide array of impacts of integrating these systems into the grid. We find two recurring analytical themes in the literature, which we use to organize the remainder of this section:

- 1. The *physical (reliability) and economic (costs and benefits) impacts* of distributed generation on the distribution, transmission, and/or generation components of the power system.
- 2. The *decision-making* and planning perspectives that mediate these impacts in the context of the electricity system components considered.





2.1 Physical and Economic Impacts

Papers examining the integration and optimization of distributed generation most often address their impact on distribution systems and primarily focus on rooftop solar generation. All of these papers fall in area 1 (Figure 2.1). The distribution system faces numerous physical challenges from rooftop solar penetration [7]. Voltage regulation is a significant area of research within models that explore various planning objectives, including those that (1) minimize curtailment with various control mechanisms [19]; (2) minimize the cost of utility power purchases [13] and DER generation cost [20]; (3) optimize the siting and sizing of distributed generation [21] and of distributed storage [22]; and (4) minimize distribution losses [23]. Other papers are more methodology-focused and review different power distribution planning models that address DER [24]. However, these studies are largely illustrative and not applied to actual planning problems with different objectives and planning constraints. At the transmission level (area 2 of Figure 2.1), research has examined voltage regulation and reactive power impacts [18], as well as transient and steady state stability [17]. It is important to note that few, if any, papers empirically reflect the planning challenges that utilities face: electricity production cost minimization, reliability requirements, and adherence to renewable procurement requirements.

Cost minimization is the most common criterion for economic assessments of DER adoption in the papers reviewed. However, the costs included in the objective functions vary among these papers. For example, some research focuses on minimizing the cost of utility power procurement. A paper minimizes the cost of procurement and self-owned DER while taking into consideration distribution and transmission system constraints (area 4 of Figure 2.1) [13]. Another paper addresses minimizing DER procurement and ancillary services costs, but limit the scope to the distribution system (area 1 of Figure 2.1) [20]. Papers also address costs in area 1 of Figure 2.1, including minimizing distribution losses and investment costs with storage [25], minimizing substation delivery and investment costs through the

optimal siting of DER [26] and determining least-cost distribution investments through active network management [27]. The aforementioned literature is largely limited in scope to the distribution system and DERs within it, without considering the impacts to the entire electric power system. A study closer to ours take a broad physical scope (area 5 of Figure 2.1) by minimizing the cost of transmission, generation, and storage [28]. The paper, however, is not empirical and does not consider the distribution system.

The economic assessments of the physical impacts of DER on the distribution system are based on objective functions that represent utility- or customer-centric interests. This suggests that the operational and investment decisions that inform these impacts are mediated by the decision-making framework, which we discuss in the following sub-section.

2.2 Decision-making Framework

We find that analyses of distributed generation investment and dispatch fit within either a centralized [13,20] or decentralized [29,30] decision-making framework. Current rooftop solar deployment is characterized by a decentralized approach to investment and management. Utility customers either own or lease the majority of rooftop solar PV installations and design and manage their systems for self-optimization subject to rate structures and levels [31]. *There are no known studies that compare the system-wide costs from centralized and decentralized decision-making frameworks under different levels of DER penetration.*

Existing research on these decision-making frameworks has generally neglected a topological representation of the grid and focused on the customer-utility relationship from an economic perspective (e.g., through rate incentives). Papers on this topic represent the interactions between customers and utilities using game-theoretic models, including Stackelberg games, which determine customers' utility-maximizing demand response and load shifting decisions under various utility pricing schemes and incentives [32–34]. A related branch of work studies the impact of different rate structures and net-metering policies on customer DER deployment. [31] finds that distributed PV deployment is highly sensitive to retail rate structure and [35] further refines how net-metering impacts retail customer's bill savings. On the storage side, [36] and [37] explore its scheduling and adoption under different rate structures, respectively. The preceding research generally shows that retail rate structures and net-metering play a critical role in future DER deployment. These findings motivate a more thorough examination of retail rates in our analysis, even if long-term planning processes do not always fully consider the direct impact of retail rates.¹

In summary, we find that the complexity of the electric grid and interdependence of its components pose some challenges for analyzing its evolution. Recent research has generally focused on a (1) subset of the entire power system (e.g., distribution versus transmission); (2) limited set of potential impacts such as voltage regulation or power purchase costs; and (3) centralized decision-making paradigm. This paper addresses these gaps by focusing more broadly on the entire electricity system (Area 7 of Figure 2.1); evaluating a range of physical and economic metrics (e.g., line losses, reliability constraints,

¹ IRP activities often include the impact of rates indirectly through price elasticities, but these are generally simplified to the response of demand to a single volumetric rate level and not to more complex (or varied) rate structures.

wholesale prices, infrastructure costs); and ultimately assessing the economic value that would result from different decision-making frameworks (e.g., centralized utility planning versus decentralized decisions by end-use customers) on an empirical representation of a U.S utility.

3. Methods

In this section, we introduce the GAP model employed to assess the differences between centralized and decentralized decision-making paradigms. The original GAP model implements a centralized decision-making paradigm for investment and operation of the power system. For this paper, the model is modified to emulate a decentralized decision-making paradigm by splitting the customer (behind the meter) decisions from the utility expansion decisions. These two model versions are then used to simulate a set of scenarios and compare the outcomes of each paradigm.

3.1 GAP Model Description

GAP is a joint capacity expansion and production cost model that determines the least-cost investment and dispatch decisions to operate a power system over a long-term time horizon, subject to numerous constraints that ensure that results are both possible and reasonable. GAP is based on the SWITCH model, an open source generation-transmission expansion model that was developed at the University of California, Berkeley and has been employed in expansion pathways studies across the world [38–41].

GAP is a deterministic linear program whose objective function is to minimize the net present value (NPV) of investment, operation, and maintenance of the generation and storage units, transmission lines, distribution network, and DER technologies in a power system (see equation (1)). The model meets residential, commercial, and industrial demand at every node in every sampled hour by dispatching the available utility-scale and DER generation and storage units and transmission and distribution lines. Utility-scale generation is split into baseload, flexible, variable, and "peaker" technologies such as natural gas-fired simple cycle combustion turbines, recognizing the operational constraints of each technology type including its capacity to provide spinning and non-spinning reserve and ramping, among others. Several "conservation" constraints assure that basic power system physical balancing is met. A detailed mathematical representation of the GAP model adapted from Carvallo et al. (2019) is included in Appendix A (see that Appendix for a nomenclature table).

 $\min(\text{total cost} = \text{NPV}(I_G + I_T + I_D + I_R + 0\&M_G + 0\&M_T + 0\&M_D + 0\&M_R + F_G + F_R))$ (1)

where NPV is the net present value function; I_{G} , I_{T} , I_{D} , and I_{R} are the annuities of generation, transmission, distribution, and DER investment, respectively; $O\&M_{G}$, $O\&M_{T}$, $O\&M_{D}$, and $O\&M_{R}$ are the non-fuel operation and maintenance costs, and F_{G} , F_{R} are fuel costs for utility scale and DER generation.

GAP was developed with the objective of assessing the tradeoff between investment and operation of distributed versus utility-scale resources. For this purpose, GAP represents generation units, transmission and distribution lines, and distributed resources with explicit temporal and spatial resolution.

Temporally, GAP uses two levels to represent operation and investment decisions. The model samples chronologically related hours—called *timepoints*—to simulate dispatch decisions on representative

peak and median load days throughout the year. The model makes investment decisions in investment periods, which usually encompass one or more years. There is a set of representative timepoints in each investment period, and the set of investment periods constitutes the simulation horizon. Each timepoint has a weight that is proportional to the number of hours in the investment period that the timepoint represents. The specific model calibration for spatial and temporal resolution used in this paper is reported in Section 4.

Spatially, GAP uses two levels of aggregation to represent the transmission and distribution systems. The bulk transmission system is divided into load zones, which represent the most important transmission nodes. Within a load zone there is no transmission congestion; the transmission of power between load zones is modeled with a transshipment or transportation model. Within each load zone there is a simplified representative distribution system consisting of dozens to hundreds of load nodes connected through an existing distribution network, whose investment and dispatch are also simulated using a transportation model. The limitations and consequences of these modeling choices are explained in Appendix B.

The GAP model simulates a basic primary distribution system expansion and operation by solving a network flow problem with variable edge or circuit capacity using a transportation model. These types of models obviate the physical complexity that AC optimal power flow (OPF) and DC OPF approaches can represent. However, the choice of a transportation model is necessary to make the model computationally tractable, especially to solve the centralized problem. Even though transportation models are simplified, research has shown that these representations are reasonable approximations for distribution and transmission network capacity expansion modeling [42,43].

It is important to note that the intended use for the GAP model is as a high-level planning tool by regulatory staff and utility planners who seek to understand the interactions between distributed and utility-scale resources over time and space. The model is not intended to produce investment decisions related to the procurement of actual resources.

3.2 Decentralized Decision-making Model Implementation

The GAP model is based on the perspective of a centralized planner who controls all decisions about investing in and operating its power system. This single planner optimizes the entire integrated system, from utility-scale power plants to transmission expansion to residential-scale DER. However, in reality, DER are typically deployed and operated behind the meter by customers following their own incentives to minimize electricity bills or maximize profits. Consequently, it is unlikely that customers would choose to implement the same DER investment and operation decisions that are featured in the least-cost integrated power system expansion. Therefore, the optimal decisions determined by the original, centralized GAP model may have limited practical value despite providing a helpful representation of an "ideal" power system. Its results do not project how customers will invest in and operate DER based on their own objectives, and it cannot shed light on optimal utility-scale decisions with respect to customer-driven DER deployment. We decompose the centralized GAP model into customer and utility problems that together define a resource planning framework with decentralized decision-making to assess the effects of potentially conflicting utility and customer objectives.

In the customer problem, customers decide between investing in and operating DER, and purchasing electricity from the utility to minimize the total net present cost of satisfying their demands for electricity. DER deployment is allowed for residential and commercial customers, but in this implementation we do not allow industrial loads to deploy these resources. The customer problem assumes that customers can purchase any amount of electricity from the utility under a known rate structure. The differences between customer electricity demands and DER generation levels in the optimal solution to the customer problem are passed to the utility problem as net loads. The utility is forced to satisfy these net loads by using its existing assets and expanding its utility-scale generation, transmission, and distribution systems. A flowchart that illustrates this solution procedure for the decentralized decision-making paradigm is presented in Figure 3.1.

In the utility problem, the utility determines the least-cost pathway for satisfying these net loads which is consistent with traditional IRP modeling processes—but it also includes distribution system expansion. Table 3.1 provides an overview of the originally centralized and the decentralized GAP model as modified for this analysis.





The customer problem uses the same availability and costs for DER as the centralized model, but also includes an electricity rate structure. Rates are specified exogenously with enough flexibility to represent hourly and node-level rates, if needed. Rates can be calibrated to existing rate structures, specified by the model user to consider alternative rate schemes, or determined via a preliminary run of the centralized model (see Section 4.2 for more detail). In the decentralized framework, the utility problem requires net loads as inputs, which are obtained from the optimal solution to the customer problem after subtracting the contributions of DER from the exogenous demand forecast.

Table 3.1 Overview of the centralized and decentralized GAP models

Model	Objective	Constraints	Key Inputs
Centralized model (original GAP model):	Minimize total net present cost of power system, inclusive of all investments and operational dispatch	Utility-scale generation constraints, DER constraints, transmission constraints, distribution constraints	Customer electricity demands; performance, cost, and initial installed capacity of utility-scale technologies, DER technologies, transmission, and distribution
Decentralized model (modified GAP model): Customer problem	Minimize total net present customer costs, inclusive of DER deployment and purchases from the utility	DER constraints	Performance, cost, and initial installed capacity of DER technologies; electricity rate structure
Decentralized model (modified GAP model): Utility problem	Minimize total net present utility costs, inclusive of utility-scale generation, transmission, and distribution, but excluding DER	Utility-scale generation constraints, transmission constraints, distribution constraints	Performance, cost, and initial installed capacity of utility-scale technologies, transmission, and distribution; customer net electricity demands from optimal solution to customer problem

The sequential modeling approach for the decentralized scenarios used in this paper implies that the customer first invests in DER under exogenous electricity rates, then the utility must satisfy net loads. The structural distinctions between this approach and an equilibrium model are important to emphasize, as they affect how the model results should be interpreted. Since the rates are exogenous, our approach is not designed to lead to an equilibrium between the marginal costs of power supplied by the utility and power produced behind the meter by customer-controlled DER. Achieving such an equilibrium would require a model that allows for endogenous determination of electricity rates to balance supply and demand. In addition, our model does not attempt to internally calculate the "right" electricity rates. Hence, there is no guarantee that the exogenously assumed rates would generate adequate revenue for the utility to recover its costs. In Section 5.3, we examine the cost recovery implications of various rate structures, which are important in a setting where customers can invest in DER to reduce their grid electricity purchases when rates are high.

On the whole, our decentralized scenarios should be interpreted as "what-if" developments of the electricity system with respect to exogenously defined rate structures. Those rates may be well calibrated, or they may lead to technical or financial problems in the future, depending on how the user defines them. Our model can project that these problems would arise under a given rate structure, but

it does not contain a mechanism to endogenously adjust rates during the course of a model run. In this sense, we believe our sequential structure in which customers invest in DER, then the utility satisfies net loads, is a reasonably accurate representation of the current decision-making environment. Especially in the short run, the retail rates facing customers are fixed, and the utility is legally obligated to satisfy their net loads. In the long run, the utility could petition its regulator to adjust rates in order to address revenue adequacy issues that might emerge, but this is not a simple process of equilibration and it is also far from exact.

3.3 Formulation of the Customer Problem

Customers are defined as the collection of either residential or commercial entities at a given node of a load zone who consume electricity, and own and operate DER technologies². We assume the behavior of customers at different nodes are independent in each period. The customers' objective function is to minimize the total net present cost of the electricity required to meet their demands with full reliability.³ There are no interactions among customers at different nodes because the customer problem assumes that the rate structure is exogenous and that the customers can procure any amount of electricity for these rates. Customers then are not directly subject to utility-scale generation, transmission, and distribution constraints. Therefore, rather than solving thousands of smaller customer problems, we combine the problems faced by customers at all nodes into a single optimization problem faced by a representative customer agent. The composition of the total cost is represented by equation (2):

total customer cost = DER costs + electricity purchases - net metering revenues (2)

First, the customer incurs DER investment and operational costs ("DER costs"). Since the customer can purchase an unlimited amount of electricity from the utility at known rates, they will only invest in DER insofar as their costs are offset by reduced electricity purchases or increased net-metering revenues. In this analysis, the customer can deploy and operate either or both of two specific DER technologies: rooftop PV panels for generation and distributed battery storage. Rooftop PV deployment is modeled with an upfront capital cost, a fixed annual operation and maintenance cost, and no fuel costs, with capital costs declining over time. Storage is an emerging technology that could significantly change the way distribution systems operate. We assume that storage technologies have decreasing capital costs over time, following [44]. Furthermore, battery storage is modeled with separate upfront capital costs for discharge capacity (MW) and for energy storage capacity (MWh). Storage operational levels are tracked across successive timepoints, subject to storage capacity and efficiency losses.

Second, the customer purchases electricity from the utility in order to complement DER generation. In our model, the customer must pay for the net electricity consumed at each timepoint. This is a simplification of typical monthly billing cycles, which would require complex tracking and balancing of monthly net demand. The total "bill" for each investment period is calculated as the sum of the

² We assume that industrial customers are connected at the transmission level and hence are not considered part of distribution system load nodes.

³ Full reliability means meeting all demand in all simulated hours. Given the temporal resolution and deterministic nature of the model, simulating perfect reliability is a conservative way to assure the results comply with resource adequacy requirements.

expenditures in each timepoint weighted by the duration that the timepoint represents (see Section 3.1 for an explanation of the temporal representation).

Third, the customer may earn revenue through a net energy metering (NEM) program, which is represented in the model as a negative cost. With net-metering, the customer can sell excess electricity back to the utility when their PV system generation exceeds end-use demand. As with the "bill" for electricity purchases, the net-metering policy in this model is calculated monthly. If the monthly DER generation exceeds the end-use demand, the excess energy will be purchased by the utility at a surplus electricity price. The surplus electricity price is typically specified in advance either as a flat rate in \$/MWh or as a percentage of a time-varying electricity rate. Customer demand must be satisfied according to the constraint specified in equation (3) below:

(3)

customer monthly demand = DER monthly generation + electricity purchases – net meteringenergy

From the utility's perspective, it is important that the rate structure be designed in a way that induces customers to make appropriate investments in DER. If rates are too low, then customers will adopt little to no DER and need to acquire more electricity from the utility. This could potentially increase stress on the utility's existing assets and force the utility to make costly investments in its utility-scale generation, transmission, and distribution systems. If rates are too high, then customers will deploy substantial DER and supply a significant amount of their own electricity. This could be a more expensive pathway due to higher per-unit capital costs at the distributed scale, and the utility might struggle to recoup its costs due to dwindling electricity sales. In either case, rates that are too high or too low can result in investments that diverge from the least-cost solution, resulting in a more expensive power system.

The customer problem in our decentralized model has several noteworthy limitations. The most important is the absence of fixed costs such as connection or capacity charges, whose inclusion would turn the customer model into a mixed-integer problem and make it computationally intractable. Second, a node represents a collection of either residential or commercial entities and hence cannot represent household- or commercial facility-level costs such as bi-directional meter installation fees or the annual fee for participating in a net-metering program. Third, our assumption that the customer can purchase any amount of electricity from the utility ignores the short-term rigidity of the generation, transmission, and distribution systems that the utility has to expand to procure that power. Fourth, to ensure that the model can be solved in a reasonable amount of time, we omit elasticity of final demand and economies of scale associated with DER deployment. This assumption could cause the model to underinvest in DER in scenarios with high penetration and decreasing costs. Finally, we assume that the rate structure and net-metering policy are independent and exogenous inputs, such that altering either one does not automatically lead to a change in the other.

4. Case Study and Input Data Creation

The GAP model has been employed previously to study conceptual systems. For this study, however, we deemed it useful to calibrate the model with input data that resemble an existing LSE's generation, transmission, distribution, and customer-sited DER assets. We use an actual utility from the Western

U.S.⁴ as a case study for this report, chosen based on its size, existing and potential DER penetration, and vertical integration⁵. These characteristics make the setting interesting, produce a model that is computationally tractable, and lead to results that are potentially applicable. *Utility* also conducts an annual long-term planning process that provides publicly available information to calibrate the supply side of the model, including load and fuel cost projections, existing plan inventory, and a discussion of potential future resources. Data sources are discussed in the subsections below.

As of 2018, *Utility* had approximately 630,000 customers, total sales of nearly 12,000 GWh, and peak demand of ~3,000 MW. *Utility*'s service territory currently has ~190 MW of distributed PV capacity, with growth in PV expected to exceed 500 MW by 2030.

The representation of *Utility* in this paper departs from the actual utility in several ways (see Table 4.1). We are confident that the omissions and adjustments made to *Utility's* real operation do not affect the scenario-based comparisons used in this analysis.

Characteristic of Utility not	Reason for Omission	Potential Impact
Represented in Implementation		
Imports: <i>Utility</i> purchases about 25% of its peak demand requirements from the wholesale market	Modeling imports would require depicting and modeling areas neighboring Utility, which are outside the scope of this paper.	Low: Market purchases are probably a mix of natural gas and solar/wind generation, which are the same resources used for capacity expansion in the Utility implementation.
Cogeneration: Utility produces about 40% of its power through cogeneration	Modeling cogeneration requires steam demand data, which are not readily available.	Medium: The natural gas plants that Utility uses for cogeneration are simulated, but possibly with higher availability and flexibility compared to real operation conditioned by steam demand.
Actual distribution system	Detailed information on Utility's distribution system is not publicly available and our efforts to obtain it were unsuccessful.	Low: We develop a reasonable approach to recreate a distribution system that should be similar to Utility's (see Section 4.1).
Actual locations of existing DER within the distribution system	The information is not publicly available.	Low: Low for a whole-system analysis, but impact could be

⁴ Representatives from a Western U.S. utility, which served as a case study for this analysis, provided constructive comments on this manuscript. Utility staff requested a desire to remain anonymous. Accordingly, for the remainder of this paper, we refer to this LSE simply as "Utility."

⁵ This paper does not seek to recreate the utility's investment recommendations from its IRP or to provide alternative expansion options for the utility. As discussed earlier, this analysis employs a substantially different modeling approach by integrating deployment decisions for utility-scale *and* distributed resources within the same model framework.

Characteristic of Utility not Represented in Implementation	Reason for Omission	Potential Impact
		higher if the actual feeder topology were employed.
Transmission system capacity	The information is not publicly available.	Low. We estimate existing transmission capacity based on voltage levels. Transmission capacity can be expanded as needed.

4.1 Input Data Creation

This subsection explains the methods, data sources, and assumptions used to create the input data for the *Utility* case study analysis. More specifically, we discuss assumptions related to the transmission and distribution networks; peak demand and model simulation; existing and new utility-scale resources; and existing and new distributed resources.

Spatial components: transmission and distribution networks

GAP represents the transmission network by grouping actual load centers into "load zones." The load zones represent nodes in the transmission system among which transmission capacity constraints and expansion decisions are modeled.

In this implementation, load zones are created by clustering buildings in the *Utility* service area. We obtain a database of buildings from the OpenStreet project [45]. Buildings are clustered into 21 load zones using a k-means clustering algorithm (see blue polygons in Figure 4.1). Transmission capacity between load zones is estimated by overlaying existing substations and transmission lines above 66 kV with the load zones. We then identify the lines that link specific load zones and aggregate those to produce a simplified transmission network (see red lines in Figure 4.1).

Unfortunately, we do not have access to any distribution network data from *Utility*. Accordingly, we implement the following procedure to produce a proxy distribution system based loosely on *Utility*⁶:

- We leverage the "taxonomy of feeders" work developed by the Pacific Northwest National Laboratory (PNNL) and commissioned by the U.S. Department of Energy [46]. In this work, PNNL analyzed 575 distribution feeders across the country and produced a taxonomy of 24 prototypical feeders associated with five climate regions.
- 2. These prototypical feeders were made publicly available in Grid Lab-D format. The information available includes poles, transformers, overhead and underground lines, customers, and operation equipment. NREL complemented this dataset with an estimation of a year (8,760 hours) of hourly load data for each customer connected to a feeder [47].
- 3. The five feeders in the climate region where *Utility* resides are used to represent its feeders.

⁶ Additional information on the techniques discussed in this section can be found in Appendix C.

Four of those five feeders have load data: a moderate suburban and light rural; a small urban center; a heavy suburban; and a light rural feeder.

4. Each load zone is assigned a representative feeder based on its building density per unit area. For example, load zones 20, 17, and 4 (Figure 4.1) are assigned light rural, urban center, and moderate suburban feeders, respectively.

Figure 4.1 Load zones and simplified transmission network we develop for the Utility service area



The feeders assigned to each load zone have varying numbers of nodes, ranging from 52 nodes in the small urban downtown feeder to 337 in the moderate suburban feeder. The final *Utility* model has 3,524 load nodes and 6,330 total nodes. In each one of these nodes, the model meets demand in every simulated hour by using a combination of DER PV, DER battery storage, and utility-scale power flowing through the transmission and distribution networks.

Temporal components: load and simulation resolution and horizon

The hourly demand at each feeder load node is scaled by a factor proportional to its building density to make the aggregate peak demand match Utility's actual peak demand. This scaling process is undertaken for two reasons. First, it ensures that load zone demand is consistent across load zones

given the lack of spatially explicit load data. This consistency is important because it matches locationspecific resources with demand requirements and allows potential mismatches in resource location and demand to drive transmission and distribution capacity investments (if needed). Second, the scaling process also ensures that aggregate distribution system demand is consistent with existing transmission and generation capacity.

The hourly load profiles in the taxonomy feeders distinguish between commercial and residential nodes. Aggregate demand is split into residential, commercial, and industrial segments based on customer segment data reported in *Utility's* 2018 Annual Report. Next, residential and commercial demands are allocated to load zones using the scaling factor, and then allocated to residential and commercial nodes using their actual peak demands. About 1,320 MW are designated as industrial demand and assigned to the head nodes of each feeder (i.e., this demand does not require use of the distribution network). Residential and commercial load grows 1.23% per year following the average annual growth reported in *Utility's* 2017 planning report. We hold industrial load growth constant over time due to a lack of information on the growth rate for this type of end-use customer.

GAP operates according to two different time steps for investment and dispatch:

- *Investment*: The simulation for this analysis is conducted for three investment periods: 2020–2024, 2025–2029, and 2030–2034. We use five-year increments—instead of annual—to ensure that the model is able to solve in a reasonable amount of time.
- *Dispatch*: GAP samples 12 chronological hours in a day (every other hour) for two days in the month: the peak demand day and a median demand day. We choose one month per quarter to capture any seasonality in demand and renewable resources. This results in 96 simulated hours per investment period, for 288 total simulated hours. Least-cost system dispatch (centralized or decentralized) is performed on each of these 288 hours at the generation, transmission, distribution, and DER levels.

Existing utility-scale resources

We extract *Utility's* existing supply-side resources from the Energy Information Administration's Forms 861 and 923 [48], in addition to *Utility's* 2018 resource planning report. A summary of *Utility's* generation mix is presented in Table 4.2.

Generation capacities listed in Table 4.2 include projects that are currently operational or have been committed with a known in-service year. Plants operating inside the *Utility* service territory are assigned to load zones based on their actual location. Plants located outside the Utility footprint are assigned to a load zone based on proximity and an appropriate interconnection cost is assessed based on their distance. The supply side mix includes three 500 MW CCGTs identified as "Market Purchases" (1,500 MW in Table 4.2) that emulate *Utility's* market purchases. We employ CCGTs because it is likely to be the marginal generation technology for market purchases.

Table 4.2 Capacity of Utility's existing generation resources

	Nameplate
Technology	Capacity (MW)

Biogas	72
Combined Cycle Gas Turbine (CCGT)	940
Geothermal	51
Hydro Reservoir	607
Hydro Run-of-River (RoR)	475
Market Purchases	1,500
PV	301
Simple Cycle Combustion Turbine (SCCT)	72
Wind	392

Monthly capacity factors for hydropower plants are estimated using historical data. Variable capacity factors for run-of-river hydro plants are estimated as the ratios between average monthly generation and nominal capacities in three years of operating data. For simplicity, these capacity factors are held constant in all future periods.

Hourly capacity factors for existing wind and solar plants are obtained from the data developed by [41] for a capacity expansion model for the Western U.S. Wind turbine output comes from the 3TIER dataset used in the Wind and Solar Integration Study [49]. Solar PV output is based on gridded solar insolation data developed by the State University of New York [50] merged with gridded weather data developed by the National Center for Environmental Prediction [51], processed with NREL's System Advisor Model. These datasets contain thousands of sites that are identified by their coordinates; we select sites close to existing *Utility* wind and solar PV farms to create input data consistent with them.

New utility-scale resources

The GAP model is supplied with a set of proposed projects from which the model selects the types of technologies and capacity size to invest in, subject to limits on some technologies that are resource constrained. The portfolio of new resources is based on the technologies suggested in *Utility's* 2018 planning report, with the following setup:

- SCCTs of 150 MW of maximum capacity are proposed in the same load zones as existing CCGTs or SCCTs.
- Wind sites are proposed only outside *Utility's* service area. *Utility's* planning documents suggests 1,200 MW of potential in CA and 900 MW in the Western U.S. (WY and NM). We include fourteen 150 MW proposed wind farms: eight in CA, four in WY, and two in NM. We calculate linear distances from each site to the utility service area and use this distance to estimate a transmission cost adder for each proposed project.
- Solar sites are proposed both inside *Utility's* service area and outside. The IRP suggests 1,000 MW of potential in domestic solar resources, 110 GW in CA, and 70 GW in the Western U.S. For simplicity, we define one project per load zone and distribute their capacities proportional to their areas.
- Battery storage projects with 150 MW maximum discharge capacity are proposed in each of the

21 load zones. The energy storage size—the number of hours of storage at full discharge capacity—is a decision variable for the model.

Capacity factors for new wind and solar PV projects are calculated following the same procedure described above for existing resources, using the appropriate locations. It is important to note that the model does not make discrete or "bulky" decisions on new project deployment. For every project, the model decides on a continuous investment in capacity per period ranging from zero to either a maximum capacity (if available) or whatever capacity is technically and economically cost-effective.

Existing and new distributed resources

Utility reports about 200 MW of existing DER PV and no DER storage in its operational area. However, there is no information on the location of these resources within the system. We address this limitation with the following procedure:

- 1. We use LBNL's Tracking the Sun dataset to estimate installed capacity by ZIP code for residential and commercial customers [52].
- 2. We develop a mapping of ZIP codes to load zones in GIS and weight each ZIP code's capacity by the fraction of its footprint that falls in each load zone. This method assumes an even distribution of PV deployment across each ZIP code. However, since ZIP code level capacity is aggregated to a larger geographic unit, this simplification will not significantly skew the estimate of load zone capacity.
- 3. Each load zone contains nodes that represent sites of residential and commercial facilities. For each node, we calculate its share of residential and commercial annual zonal load. Then, by assuming that PV deployment scales directly with load, we use the shares to weight load zone-level residential and commercial installed capacity to estimate node-level capacity.

For new DER, the model can install PV and battery storage DER on any of the 6,330 nodes in the system, including adding DER PV to nodes with existing capacity. There is no *a priori* limitation on DER deployment such as a hosting capacity limit for feeders, because the utility can adapt to optimal DER deployment by expanding its network.

Finally, we develop capacity factors for residential-sited and commercial-sited PV in each load zone. Using the same insolation input data for the utility scale—but limited to insolation inside the utility's footprint—we use NREL's System Advisor Model to estimate PV generation for fixed-roof systems for 14 cities in *Utility's* territory that fall within or overlap with each of our 20 load zones. The hourly generation is divided by the system DC capacity to create annual capacity factors.

Technology costs

GAP model results depend on the assumed capital costs of DER and generation technologies, their fuel, non-fuel O&M, and fixed costs, and expansion costs of the transmission and distribution networks. See Appendix B for a detailed account of the cost assumptions used in this analysis. In this subsection, we highlight the capital costs for key technologies, fuels, and infrastructure. All costs are expressed in real 2018 dollars unless otherwise noted.

Renewable resources including PV and wind are expected to continue their capital cost reduction trends from the last decade [53]. The pace and extent of these reductions will have an important impact on decisions that are based on least-cost dispatch. We use data from NREL's 2018 Annual Technology Baseline (ATB) report, which documents a realistic set of input assumptions and their evolution (e.g., capital costs, fuel costs) to inform electricity sector analysis in the U.S. [53]. Based on this data, utility-scale PV costs are expected to decline from 0.92 \$/W in 2020 to 0.71 \$/W in 2035 and rooftop PV is expected to decline from 2.01 \$/W in 2020 to 1.32 \$/W in 2035. In addition, onshore wind power costs are expected to decrease from 1.67 \$/W in 2020 to 1.33 \$/W in 2035.

The evolution of battery storage costs is particularly hard to predict given the relative immaturity of this technology in the market. The ATB relies on a study by [44], which projects that utility-scale battery storage costs will decrease from 488 \$/MWh at present to 192 \$/MWh by 2035. There is no known information on future capital costs of distributed storage. However, the modular nature of batteries suggests that there may not be substantial economies of scale for utility-scale deployments when compared to widespread adoption at the distributed scale. We assume a 10% cost adder for DER storage and use a cost of 537 \$/MWh at present decreasing to 211 \$/MWh by 2035.

The most important fuel cost assumption for the *Utility's* case study is natural gas. We use two sources of information to define a base case price forecast for natural gas: (1) the projections included in *Utility's* 2018 planning reports and (2) an average of Western U.S. LSEs' base case or reference natural gas forecasts extracted from their latest IRPs through the Berkeley Lab's Resource Planning Portal.⁷ *Utility's* forecast was roughly 10–15% lower than the average of the other forecasts, with no obvious explanation for this difference. We decided to use the average of Western U.S. IRPs as the natural gas fuel price as a more conservative parameter.

Transmission and distribution system expansion costs are simplified and normalized using a single dollar per capacity-length value, expressed in \$/MW-km. Transmission costs are assessed at 5,000 \$/MW-km based on an estimation of length and capacity of *Utility's* transmission lines and the infrastructure cost reported in its annual report. Distribution costs are assessed using a similar approach and yield 25,000 \$/MW-km. It is important to note that the value of existing infrastructure is similar to the actual investment value reported by *Utility*. However, these average values may be lower than the marginal costs of expanding both systems, given the impacts of interrupting service and the technical complexities of expanding distribution and transmission circuits and substations, especially in urban areas. This caveat is particularly relevant for distribution systems, whose expansion costs may affect DER adoption directly. We address this limitation by running a sensitivity centralized decision-making scenario with distribution costs ten times higher based on a distribution expansion analysis to accommodate DER [2].

4.2 Scenarios

We analyze several scenarios to study the impact of centralized versus decentralized decision-making paradigms on expansion strategies and the deployment of DER. The scenarios are used to model different decentralized decision-making problems, because their outcomes depend on the rates and

⁷ See <u>http://resourceplanning.lbl.gov</u>.

incentives available to customers. It follows that a thorough analysis requires simulating different rate structures and levels. For simplicity, we perform a single run for the centralized model, which is used as a benchmark to evaluate the optimal integrated solution from a social planner perspective. Then, we consider three decentralized decision-making cases:

- A series of flat volumetric rate scenarios to assess how the balance between utility-scale and distributed generation changes as the rate varies. We use arbitrary rates of 10, 12, 14, 16, and 20 ¢/kWh, which span most rate levels offered by U.S. electric utilities. For simplicity, we report results for the extreme flat rates of 10 and 20 ¢/kWh, and the median rate at 14 ¢/kWh.
- 2. A time-of-use (TOU) rate structure based on the one currently implemented by Utility. This rate has summer and non-summer seasons, with three pricing blocks in the former and two in the latter (see details in Table B-4 in Appendix B).
- 3. A NEM policy scenario, in which consumption and DER generation are net-metered monthly using a flat retail rate structure.

In the final core scenario, we evaluate how a utility could strategically establish its rate structure to induce customers to adopt DER at levels consistent with those in the true, system-wide optimal solution from the centralized model. This is accomplished by the utility setting its time- and node-specific rates equal to the shadow prices of the supply and demand balance constraints from the centralized model run. It follows that these shadow prices reflect the full marginal costs of satisfying demand across time and space. Table 4.3 summarizes the scenarios considered in this analysis.

Finally, three sensitivity scenarios are tested to better understand the impacts of key variables: a low DER PV cost scenario (10% reduction compared to BAU), a low DER storage scenario (15% reduction compared to BAU), and a centralized scenario with higher distribution expansion costs (ten times more expensive than BAU). The first two sensitivity scenarios are applied to both the centralized scenario and the decentralized TOU scenario. The higher distribution expansion costs sensitivity is only applied to the centralized BAU scenario.

Table 4.3 High-level summary of scenarios

Scenario Procedure	Scheme Description	Rate Structure	Scenario Name
Centralized	Single decision maker controls all	No concept of rates under the centralized model	Centralized
	decisions as a social planner.	scheme	
Customer → Utility	Customer optimizes DER under	Flat Rate: single exogenous volumetric rate for all	Decentr Flat rate 10 ¢/kWh
	exogenous rates; then utility must	nodes and times (including sensitivity analysis on this	Decentr Flat rate 12 ¢/kWh
	satisfy net loads.	rate)	Decentr Flat rate 14 ¢/kWh
			Decentr Flat rate 16 ¢/kWh
			Decentr Flat rate 20 ¢/kWh
		TOU Rates: apply the actual rate structure currently	Decentralized TOU
		implemented by Utility	
		Flat Rate with NEM: same as Flat Rate, but with net-	Decentr Flat NM 10 ¢/kWh
		metering implemented in the customer problem	Decentr Flat NM 12 ¢/kWh
			Decentr Flat NM 14 ¢/kWh
			Decentr Flat NM 16 ¢/kWh
			Decentr Flat NM 20 ¢/kWh
Centralized →	The additional first centralized step	Marginal Cost Rates: the utility applies rates that are	Decentralized Marg
Customer → Utility	allows the utility to apply rates set to	strategically set equal to the shadow prices of the	
	steer the customers toward DER	supply and demand balance constraints in the	
	decisions that are close to those in the	preliminary centralized model run, to steer	
	optimal solution to the centralized model.	customers toward optimal DER decisions that are aligned with the true least-cost power system	

5. Results Analysis and Discussion

This results section is organized in the following manner. First, we report how centralized and decentralized decision-making outcomes differ in technology and location choices as well as operational decisions. Second, we detail the differences in costs between scenarios. Third, we show the impact of different rate structures on DER adoption for both decision-making paradigms. Finally, given cost-effective DER, we introduce the concept of *price elasticity of net load* as a metric that could complement the traditional metric of price elasticity of final electricity demand.

5.1 Technology Choices

This subsection reports the differences between generation technology investment and dispatch decisions under centralized and decentralized decision-making paradigms. These differences arise in the balance between utility-scale and distributed resources, as well as the composition of the utility-scale capacity and generation mixes.

Figure 5.1 illustrates the evolving total installed capacity mix in the centralized scenario, and it shows the differences between this scenario and selected decentralized cases. Centralized decision-making would generally lead to substantially lower DER PV adoption, but higher DER storage adoption, compared to the decentralized scenarios. This is not always true, however, for distributed storage, which undergoes a spike in adoption in 2030 under the 14 ¢/kWh flat rate and TOU decentralized scenarios. There is evidence that high electricity price signals and expected capital cost declines for battery storage are sufficient to induce strong residential battery uptake by 2030 in decentralized scenarios. The customer makes these behind-the-meter decisions to minimize their electricity bill, but these decisions may not be optimal at the system level.



Figure 5.1 The evolving total installed capacity mix in the centralized scenario and differences that emerge in selected decentralized scenarios

In almost all decentralized scenarios, the utility responds to behind-the-meter DER decisions by deploying less utility-scale solar compared to the fully centralized solution. The tendency for distributed and utility-scale PV to substitute for one another has important implications for utility planning to satisfy renewable resource standards and mitigate the environmental impacts of electricity. Interestingly, distributed storage is the preferred mode of storage investment in the centralized solution, which contrasts with PV generation where the utility-scale mode is preferred. This likely reflects the smaller differences in relative capital cost between distributed and utility-scale battery storage, and the ability of DER storage to defer investments in the distribution network. Consistent with the logic behind their construction, the long-run marginal cost rates result in decentralized investment decisions being much more similar to those featured in the centralized solution, as evidenced by the smaller differences for this case in Figure 5.1.

Decentralized model results reflect the expected decline in distributed PV capital cost over time. Noticeable differences appear only in period 2030, but higher rates stimulate higher adoption to period 2025, as more expensive utility-scale electricity makes DER PV investment attractive at a higher capital cost that will be reached earlier. For this case study, a flat rate of 14 ¢/kWh⁸ drives 1.2 GW of DER PV adoption by 2025 and 3.1 GW by 2030 without a net-metering policy. The relative importance of DER PV costs and electricity rates for PV adoption is more evident in the decentralized scenario with *Utility's* TOU rates. The average annual mid-day rate—coincident with peak PV output—is about 12 ¢/kWh, whereas the peak rate of 25 ¢/kWh takes effect at 5pm when PV output is much lower. DER PV costs falling below 1.5 \$/W by 2030 are necessary to spur behind-the-meter adoption.

Compared to the scenarios without net-metering, a monthly NEM scenario completely eliminates investments in distributed storage. Since high PV production in peak hours can compensate for hours with low PV production, distributed storage is much less valuable for customers if a monthly netmetering balance is available. Similarly, the availability of NEM drives earlier and larger adoption of DER PV compared to scenarios where it is not available. While DER PV adoption is contingent on rate levels, for a given rate level a NEM policy can drive two to three times more DER PV adoption compared to a non-NEM case. This is in part due to customers investing in DER storage in non-NEM cases, and also due to the economic incentives provided by the NEM policy.

The shares of distributed and utility-scale generation resources in the overall generation mix are reported in Figure 5.2. This representation captures only production technologies (no storage). We find that substantial customer PV adoption occurs in many of the decentralized scenarios as DER PV becomes more affordable. Increasing DER PV adoption proportionally reduces utility net loads and displaces utility-scale generation. Approximately 10–15% of demand is met using DER PV by 2030 with lower flat electricity rates, which increases to 40% with higher flat electricity rates in the 20 ¢/kWh range.

⁸ A reminder that dollar values are in \$2018. A 14 ¢/kWh rate corresponds to a nominal cost of 16.4 ¢/kWh in 2025 and 18.4 ¢/kWh in 2030 assuming a 2.3% average annual inflation rate.



Figure 5.2 Balance between utility-scale and distributed generation resources in the overall generation mix

Three sensitivity scenarios are tested to better understand the impacts of key variables: a low DER PV cost scenario (10% reduction compared to BAU), a low DER storage scenario (15% reduction compared to BAU), and a centralized scenario with higher distribution expansion costs (ten times more expensive than BAU). Results show that lower DER PV costs have minimal impact on its adoption in the centralized scenario (Figure B-2, SI). This suggests that very aggressive cost reductions would be needed to make DER PV adoption cost-effective under a centralized decision-making paradigm. In contrast, lower DER storage costs would drive about 100 MW more adoption of this resource in the centralized scenario, enabling the adoption of about 150-200 additional MW of utility-scale solar PV and displacing wind capacity (Figure B-2, SI).

In the sensitivity results for the decentralized TOU scenario, lower PV costs trigger earlier adoption of the resource of ~1.5 GW in 2025 and an additional 0.7 GW in 2030 for a ~3.5 GW total in this period (Figure B-3, SI). Lower PV costs also drive adoption of about 0.5 GW additional DER storage compared to the original decentralized TOU scenario. In contrast, lower DER storage costs do not affect capacity deployment decisions in 2020 and 2025, but would trigger about 300 MW of DER storage and DER PV adoption in 2030. In general, the lower DER costs have modest impact on resource adoption across scenarios, with the exception of lower DER PV costs in the TOU scenario. Finally, higher distribution expansion costs do not impact resource adoption decisions, since the centralized model is already choosing a minimal amount of DER as cost-effective. The logic is that higher distribution costs may drive higher investment in DER as a non-wires alternative, but the low load growth for *Utility* makes this driver irrelevant.

In the decentralized scenarios, *Utility* cannot take advantage of the lower utility-scale PV costs that are correlated with DER PV cost declines, because the decentralized model follows a specific decision sequence with the customer deciding first and the utility following. The value of utility-scale PV drops with DER PV adoption given the high temporal correlation of their outputs. This may produce substantial cost inefficiencies, which we examine in the next subsection.

5.2 Economic Value of DER Integration

The first measure of economic value is calculated by comparing the costs of the decentralized solutions against the centralized decision-making optimum, viewed as a benchmark for comparison. Figure 5.3 reports how much less expensive the centralized scenario is compared to the decentralized scenarios, by period. In other words, the figure reports the percent savings of the total power system costs that would accrue should a centralized decision-making framework be adopted. We use this reporting scheme because the centralized scenarios are the "business-as-usual," whereas the centralized scenario is the counterfactual.

Figure 5.3 Percentage decrease in total period costs in the centralized scenario relative to decentralized scenarios



The TOU decentralized scenario is constructed to reflect the case study's current residential rate structure. This scenario has very similar costs compared to the centralized scenario in the first two periods, but there are significant savings by the centralized approach in the last period (-15% cost difference). This cost difference occurs because DER PV and battery storage costs drop enough to become attractive to customers by 2030, and hence are heavily adopted in that period (see Figure 5.1). However, the same renewable resource cost trend makes utility-scale wind and PV more cost-effective over time compared to other distributed and utility-scale technologies. The centralized model serves demand more cost-effectively using utility-scale resources instead of DER PV. The centralized model has an NPV of about 7% less than the decentralized model with a TOU retail rate and up to 30% less than the highest flat rate cases (Table 5.1).

Table 5.1 Difference in net present value (NP)	/) of centralized scenario a	gainst decentralized scenarios
--	------------------------------	--------------------------------

Scenario	Centralized Scenario Difference in NPV (%)
Decentr Flat rate 10¢/kWh	-10%
Decentr Flat rate 12¢/kWh	-12%

Decentr Flat rate 14¢/kWh	-17%
Decentr Flat rate 16¢/kWh	-24%
Decentr Flat rate 20¢/kWh	-30%
Decentr Flat NM 10¢/kWh	-18%
Decentr Flat NM 12¢/kWh	-23%
Decentr Flat NM 14¢/kWh	-27%
Decentr Flat NM 16¢/kWh	-32%
Decentr Flat NM 20¢/kWh	-37%
Decentralized Marg	1%
Decentralized TOU	-7%

The cost structures of both paradigms are consistent with their differences in investment choices. By 2030, the decentralized model produces a utility that has a much smaller distribution system because around one third of power is generated behind the meter. The TOU scenario does not significantly expand the distribution system. However, the centralized scenario expands the capacity of the distribution system by 50% to accommodate the native demand growth with little to no DER adoption.

Among the flat rate decentralized scenarios, we choose the 10 ¢/kWh case to compare against the centralized scenario because this is closer to the levelized cost of electricity in the centralized solution. The centralized solution has 7–19% lower costs than this decentralized flat rate. One explanation for this difference is that the centralized model deploys DER battery storage, whereas the decentralized model deploys utility-scale storage (see Figure 5.1). This is because it is generally not attractive for flat rate customers to adopt storage, particularly with a low electricity rate of 10 ¢/kWh. Since the customers do not adopt much DER storage, the utility plans around this by deploying less utility-scale PV and more wind and natural gas. By tailoring its utility-scale investments to the suboptimal (from the utility perspective) DER deployment decisions implemented by customers, the utility can only partially offset the loss of economic value stemming from behind-the-meter DER investments that diverge from those in the centralized solution for the whole power system.

The marginal cost decentralized scenario behaves as expected, with minimal cost differences compared to the centralized decision-making paradigm. In the marginal cost scenario, the customer faces rates that properly reflect the long-term marginal cost of the utility supplying power to that node on an hourly basis. It is worth noting that the decentralized solution based on marginal cost rates does not exactly match the results of the centralized scenario. Slight differences arise because the marginal costs computed during the preliminary run of the centralized model do not remain precisely accurate once the customer problem adds DER at some of the nodes. Essentially, the addition of DER can alter the marginal cost of supplying utility-scale power to each node at each timepoint, but the differences are minor enough that the marginal cost rates from the preliminary model run still induce a decentralized solution that comes quite close to the centralized optimum. In principle, TOU and critical peak pricing rate structures are implemented to at least roughly capture the temporal differences in the marginal cost of providing electricity at different times. However, it should be noted the marginal cost are such that customers would face a price exceeding 200 ¢/kWh on peak hours of the year, a rate that marginal cost set and that the marginal cost set and the term of the cancel and the term of the marginal cost of the term of the marginal cost of providing electricity at different times. However, it should be noted the marginal cost are such that customers would face a price exceeding 200 ¢/kWh on peak hours of the year, a rate that marginal cost head to the term of the year, a rate that marginal cost term of the term of term of the term of the term of term o

be politically challenging to impose.

Finally, we develop a back of the envelope calculation to extrapolate these results to all DER PV installed across the U.S. This estimate involves first calculating the difference of DER PV capacity deployment and total system cost between the centralized and decentralized scenarios. Second, we divide these values to estimate cost savings per unit of DER PV capacity. Applying this ratio to existing U.S. DER PV capacity gives a rough estimate of the costs that could have been saved had a centralized framework guided investment and operation of these resources. Based on this simple approach, we find savings of \$0.5 to \$1.6 per watt of DER PV installed, which is 25% to 75% of current DER PV costs. We used data from the EIA's Electric Power Monthly [54] to estimate \$500 million to \$2.2 billion in annual savings nationally had existing DERs been centrally dispatched. This national-level estimate is relatively small compared to current utility revenues, because of relatively modest existing DER adoption in the U.S. It can be inferred that additional market penetration of DER technologies—as is expected—will lead to much larger costs than what is reported here unless this deployment follows a centralized decision making paradigm.

5.3 Price Elasticity of Net Load

The literature that focuses on how resource planning is addressing DER adoption finds that some utilities are devoting considerable effort to refining forecasts of distributed resource adoption (e.g. [4]). Key deployment drivers considered in forecasts include public policy, customer preferences, macroeconomic drivers, and DER economics. The latter driver includes DER capital costs, incentives, electricity prices, and rate design.

However, treating DER adoption as an exogenously forecasted input or consideration through scenariobased analysis misses an important causal linkage between utility electricity rates and DER adoption. To evaluate this causal relationship, we introduce the concept of the *price elasticity of net load* and use the GAP model case study to demonstrate its application. The price elasticity of net load connects the electricity rate to the net load the utility serves by capturing the effect of the rate on DER adoption and the effect of DER adoption on customers' demand for electricity supplied by the utility. As a consequence of decentralized DER decision-making, even if the final demands for electricity are fixed (as is the case in our modeling exercise), the residential and commercial demand served by the utility (rather than met using DER) depends on the rate structure due to behind-the-meter DER adoption that is rate-sensitive.

Figure 5.4 shows how the net energy demand (top) and net peak demand (bottom) vary with the flat electricity rate in each modeled period, for residential and commercial loads. Industrial loads are not included because DER deployment for this type of customer is restricted in the model, and hence their adoption is perfectly inelastic to rate levels by design. The net energy demand is calculated by subtracting the DER generation and net storage from the native demand, for each hour, and summing the weighted hourly samples to produce total annual net energy demand. The net peak demand is calculated by determining the hourly net energy demand for the whole system, and then selecting the maximum value in each investment period. We simulate the system expansion for a decentralized scenario using flat rates at 10, 12, 14, 16, and 20 ¢/kWh and then plot the net energy and net peak demand for each simulation-period combination. This method allows estimating a long-run price

elasticity of net demand, as both variable and capital costs respond to the price changes. The simulation does not suffer from the identification issues in empirical studies because the variation in rate is entirely exogenous [55].



Figure 5.4 Net energy demand and net peak demand by period as a function of the flat rate level

The slope of the annual net energy demand decreases at an increasing rate over time, most likely as a reflection of the decreasing capital costs of DER. In 2020, net energy demand is roughly 98% of the native demand for lower rates and it decreases to 77% of native demand with a 20 ¢/kWh rate. However, by 2030 even the lowest rate produces a system whose net energy demand is 81% of the native demand, declining to a plateauing level of 29% at 16 ¢/kWh. The 2025 period serves as a good example of how elastic net energy demand is to changes in flat rates: net energy demand for residential and commercial sales decreases from 10.1 TWh at 10 ¢/kWh to 4.6 TWh at 20 ¢/kWh, a 54% reduction. Using the 2025 period, which does not have the boundary issues of periods 2020 and 2030, we use ordinary least squares (OLS) regression and find a price elasticity of net energy demand of -1.05. In other words, a 1% increase in retail rates correlates with a 1.05% decrease in net energy demand. This value is relevant for two reasons. First, it is similar to other values for long-run elasticities reported in the literature that lie in the -0.7 to -1.0 range [56–58]. Second, net demand is price elastic (elasticities are elastic above 1 in absolute value) and higher than prevailing studies on native demand that are

largely based on historical billing data with little DER adoption. The elastic demand may indicate that higher levels of DER adoption could have a larger than expected effect on net demand.

The dynamic for net peak demand is very different from that for net energy demand (Figure 5.4, bottom panel). Net peak demand decreases by around 10% across the range of flat rates studied, from 2.1 GW at 10 ¢/kWh to about 1.6 GW at 20 ¢/kWh for the period 2025. The changes in net peak demand across prices follow a similar pattern as the net energy demand, but with a much smaller overall effect. The main explanation for this result is that the price signal that customers receive in this scenario—a flat volumetric rate—does not reflect the true costs of peak demand. This flat rate does not induce DER investment decisions targeted to reduce peak demand against price results in a -0.35 value, three times smaller than the elasticity of net energy demand. There are two additional explanations for the lower elasticity of net peak demand. First, peak demand for the *Utility* case study occurs in the evening hours when the value of DER PV is relatively low compared to its value during high insolation hours in the middle of the day. Second, in the absence of widespread DER storage or net-metering policies, customers have little incentive to oversize their systems to shave higher portions of peak demand.

The price elasticity of net load is relevant for utility resource planning because it connects electricity prices to utility net loads via the mechanism of customer-driven DER investment. The typical demand forecasts rarely account for the impact that higher rates have on DER adoption and, by extension, on net energy or peak demand. These results suggest that this shortcoming of traditional methods for incorporating DER into resource planning will become more problematic and pressing as DER costs decline and adoption expands. In addition, the contrasting elasticities of net energy demand and net peak demand suggest that widespread DER adoption may have a large impact on utility revenue recovery, but not on utility resource adequacy.

We use the decentralized scenario results to estimate utility revenue recovery and revenue requirements (i.e., utility costs). Revenue requirements are commonly used in regulatory planning frameworks as the metric to choose a preferred least-cost, risk-managed portfolio. Figure 5.5 shows how the utility revenue requirements and recovery change across the flat retail rate range of 10 ¢/kWh to 20 ¢/kWh over the entire simulation horizon. The utility cost or revenue requirement decreases as volumetric retail rates increase, following the same pattern as net demand shown in Figure 5.4. As customers adopt more DER, the utility incurs lower capital costs and fuel costs for utility-scale generation. Utility costs or revenue requirements decrease from \$25 billion at the lowest rate to \$22 billion at the highest rate. Simultaneously, the reduction in net demand translates to changes in physical sales, whose revenue outcome is mediated by the retail rate. In this case study, revenue increases only 50% with a doubling in the flat rate. This is because, as seen in the bottom panel in Figure 5.5, residential and commercial customers substitute grid power with DER power, reducing their contribution to utility revenue as rates increase.

The aggregate results mask the important inter-temporal differences that occur due to higher uptake of DER in the last period of the simulation. For all the flat retail rate levels, the utility revenue recovery is

insufficient to cover utility costs in the 2030 period. The shortfall is approximately 10–20% at the lower retail rate levels but increases to 30–40% at the higher rate levels.



Figure 5.5 Utility cost and revenue for decentralized flat rate scenarios

These results have important implications for utilities that are not vertically integrated. A distributiononly utility will lose sales due to declining net demand, but it will not be able to internalize the cost reductions that are driven upstream in the generation segment. Conversely, utilities that are vertically integrated should be able to balance costs internally between the distribution and generation segments, although this is contingent on the regulatory framework to set rates and determine revenue recovery. In all cases, however, vertical integration may not be enough for utilities to be financially sustainable in cases where DER costs decrease in a manner similar to what was assumed for this study.

6. Conclusion

Electric utilities have traditionally relied on least-cost capacity expansion models with a single decision maker to support their resource planning processes and identify the most cost-effective generation and transmission investments to meet expected load. However, this approach is not well suited to incorporate behind-the-meter distributed resources, which in reality are installed and operated by customers attempting to minimize their own electricity bills or maximize profits rather than make the distributed energy resource decisions that are optimal for the power system as a whole. As the costs of distributed resources continue to decline and more customers adopt these technologies, it is important

for utilities to understand the impacts of their own decisions on distributed resource adoption, and to tailor their utility-scale investments to DER deployment scenarios that customers are likely to pursue. Simultaneously, it is important for regulators to set rates that more closely approximate the true long-run marginal cost for the system, in order to drive customers to make socially optimal distributed resource adoption decisions [59].

6.1 Summary

In this paper, we adapted the GAP model for integrated generation, transmission, and distribution system expansion to analyze scenarios with a decentralized decision-making paradigm. This was accomplished by splitting the centralized decision-maker GAP model into a joint decision-making paradigm: a customer problem and a utility problem, which are solved sequentially in that order. In the customer problem, a representative customer agent minimizes the cost of satisfying all customer electricity demands by purchasing electricity from the utility at known rates and investing in distributed PV and battery storage. The electricity purchased from the utility—determined from the optimal solution to the customer problem—is used as an input to the utility problem, where these customer purchases define the net or remaining load that the utility must satisfy. Finally, the utility determines the least-cost generation, transmission, and distribution investment and dispatch decisions to supply the power demanded by customers. We apply this analysis technique to a case study based on a real utility from the Western U.S. (*"Utility"*), which has significant existing and projected residential PV adoption. We determined the least-cost system expansion using the original, centralized decision-maker GAP model and then compared that result to a number of decentralized decision-making scenarios based on alternative electricity rate structures and policy incentives (e.g., net-metering).

Several findings are relevant to utility planning processes and how these processes might need to evolve under a high DER penetration future. First, the cost reductions resulting from centralized decision-making can be substantial. In the Utility case study, the centralized solution that optimizes the entire integrated power system leads to a 23% cost reduction in 2030 compared to the prevailing timeof-use rate structure. It follows that integrating distributed resources into planning processes has significant potential economic benefits. Second, the centralized scenario generally resulted in reduced investment in distributed PV and increased investment in distributed storage compared to the decentralized decision-making solutions. In other words, the least-cost, centralized power system favors utility-scale PV and distributed storage, but behind-the-meter, customer-driven decisions tend to emphasize the opposite. Furthermore, increases in distributed PV adoption tend to displace utilityscale PV investment. The relationship of distributed and utility-scale PV as substitutes has important implications for utilities who are attempting to meet renewable resource requirements or mitigate environmental impacts. This relationship and substitutability are not currently considered in resource planning. Third, the choice of electricity rate structure is a key determinant of distributed resource adoption under the decentralized decision-making paradigm, and thus also the net loads ultimately served by the utility. We introduced the concept of price elasticity of net load to capture this causal chain, which connects rates to electricity demand. Accordingly, it will be increasingly important for utilities to understand the effects of their rates on net loads in order to make appropriately sized investments and ensure adequate revenue. Current planning practices do not generally consider the use of retail rates as a tool for bulk power system planning, even for utilities that are vertically

integrated. Finally, we demonstrated how a utility could largely overcome the complications of decentralized distributed resource decision-making by supporting and asking regulators to consider electricity rates equal to time- and location-specific, long-run marginal costs. This approach induces customers to implement distributed investments and dispatch them in a way that is aligned with the overall least-cost pathway under the centralized decision-making paradigm. While these rates may not be feasible to implement at present, the benefits to be gained by transitioning to this rate-setting approach will rise over time.

A case study-based analysis can provide useful generalizations, within the limits of the case study. We discuss the potential generalization of results for the distribution-transmission-generation components, and then the behind-the-meter component. While regulatory framework vary widely across states and countries, the basic functions of the distribution utility are the same, including the jurisdictions with retail choice. In contrast, transmission and generation investment decisions in reality depend heavily on the industrial organization, this is, whether these components are vertically integrated or not. However, both vertically and non-vertically integrated frameworks strive to produce a least-cost and timely set of investment decisions, which is exactly what our model does. Hence, we believe that the relative decisions in investment on distribution, generation, and transmission are well represented in our model and can travel well to jurisdictions with or without vertical integration. It is important to remember that a model like GAP is best used in a "what-if" approach, by comparing a base scenario with multiple variations to understand the relative impact of specific parameters. This is exactly what our model, results, and discussion do, and hence these results should apply well to interconnected power systems across the globe.

Behind-the-meter decisions will vary substantially based on the rate structures and incentives that customer face. However, the rational decision making mindset of a customer that wants to minimize its bills is a generalizable assumption that is implemented in the decentralized scenarios in this paper. Our model implements a plethora of rate structures ranging from traditional flat rates to complex real-time rates (i.e. the long-run marginal cost scenario), to make the results travel to most jurisdictions. This suggests that the set of scenarios analyzed and their results should be applicable to most states and countries around the world where customers face a regulated rate and utilities bill and collect revenue based on those same rates. We believe these assumptions and conditions make the behind-the-meter decisions also generalizable across jurisdictions.

6.2 Future Research Needs

This study is the first attempt to compare centralized and decentralized decision-making paradigms and several aspects of the model and method could be improved.

First, model optimization improvements could be undertaken to set rates exactly equal to long-run marginal costs by making them decision variables chosen by the utility. This alternative formulation is essentially a leader-follower game, where the utility explicitly selects its own rates to minimize overall system costs while anticipating the effects of its rates on customer investment on distributed resources. However, this approach is more computationally challenging than the method implemented in this study.

Another avenue for future research would involve incorporating transmission and distribution infrastructure retirements into the model, which are not currently represented in the modeling framework. As transmission and distribution infrastructure comes to the end of its useful lifespan, it is important to reconsider the relative economics of investing in the utility-scale bulk power system versus behind-the-meter distributed resources.

This study was limited to analyzing the investment and dispatch decisions using the same decisionmaking model. However, it is possible that consumers will invest in distributed resources and then outsource the dispatch and management of these resources to an aggregator. It is also possible that a third party owns the infrastructure and charges a fee to the consumer, who in-turn dispatches the resource. In the first alternative, the consumer owns the asset(s), but another operator—potentially following a centralized approach—makes the dispatch decisions. In the second alternative, a third party—potentially the utility—owns the resources and lets the consumer make the dispatch decision. These two alternatives could be studied with modest adjustments to the modeling framework.

More generally, further research could explore the capabilities of different rate and incentive structures to induce customers to make distributed resource decisions that are more closely aligned with the true least-cost expansion of the entire electricity system. Setting rates equal to long-run marginal costs is a very effective way to achieve this in principle, but it may not be feasible to implement in reality. Therefore, extensions of this research could consider a broader range of rate structures (e.g., time-of-use, critical peak pricing, tiered rates, demand charges) to compare their potential to properly guide customers toward more optimal resource choices.

Finally, this analysis showed that the centralized solution included less distributed PV deployment, but more distributed storage deployment, relative to the decentralized outcomes. This finding suggests that additional research could evaluate incentives to encourage greater customer uptake of behind-the-meter storage systems. The price elasticity of net load concept shows how the availability of economically competitive DER makes net demands—and therefore, utility revenues—more sensitive to electricity rates. Future work should explore new business models to ensure that utilities can generate sufficient revenues to cover their costs while incentivizing socially-optimal levels of distributed resource adoption.

Acknowledgements

We are grateful for constructive comments provided by a group of technical staff at a Western U.S. utility, which requested anonymity. This Western U.S. utility served as an important case study for this analysis and allowed us to refine the model in ways that more accurately represented utility operation and planning decisions.

The authors would like to Joe Paladino (U.S. Department of Energy) for supporting this research and Larry Mansueti (formerly of the U.S. Department of Energy) for encouraging us to undertake this project in the first place. The authors would also like to thank Wesley Cole, Michael Craig, Steve Rose, Fariba Mamaghani, Metin Cakanyildirim, and Sauleh Siddiqui for their thoughtful comments and suggestions. We would like to thanks Natalie Mims Frick and Naim Dargouth for their suggestions and feedback on earlier drafts. All errors are responsibility of the authors.

The work described in this study was funded by the Transmission Planning and Technical Assistance Division of the U.S. Department of Energy's Office of Electricity under Lawrence Berkeley National Laboratory Contract No. DE-AC02-05CH11231.

7. References

- [1] Paliwal P, Patidar NP, Nema RK. Planning of grid integrated distributed generators: A review of technology, objectives and techniques. Renewable and Sustainable Energy Reviews 2014;40:557–70. https://doi.org/10.1016/j.rser.2014.07.200.
- [2] Horowitz KAW, Ding F, Mather BA, Palmintier BS. The Cost of Distribution System Upgrades to Accommodate Increasing Penetrations of Distributed Photovoltaic Systems on Real Feeders in the United States. 2018. https://doi.org/10.2172/1432760.
- [3] Silva AML da, Nascimento LC, Rosa MA da, Issicaba D, Lopes JAP. Distributed Energy Resources Impact on Distribution System Reliability Under Load Transfer Restrictions. IEEE Transactions on Smart Grid 2012;3:2048–55. https://doi.org/10.1109/TSG.2012.2190997.
- [4] Mills AD, Barbose GL, Seel J, Dong C, Mai T, Sigrin B, et al. Planning for a Distributed Disruption: Innovative Practices for Incorporating Distributed Solar into Utility Planning. 2016.
- [5] Barbose G, Satchwell A. Utility-Owned Residential Rooftop Solar: An Analysis of Shareholder and Ratepayer Impacts. Berkeley, CA: Lawrence Berkeley National Laboratory; 2019.
- [6] Barbose G, Darghouth NR, Weaver S, Feldman D, Margolis R, Wiser R. Tracking US photovoltaic system prices 1998–2012: a rapidly changing market. Progress in Photovoltaics: Research and Applications 2015;23:692–704. https://doi.org/10.1002/pip.2482.
- [7] Feldman D, Hoskins J, Margolis R. Q4 2017/Q1 2018 Solar Industry Update. Golden, CO: National Renewable Energy Laboratory; 2018.
- [8] Lazard. Lazard's Levelized Cost of Storage Analysis—Version 4.0. 2018.
- [9] MarketWatch. Rooftop Solar PV Market 2019 grow at +15% and projected to grow double in upcoming years by in-depth analysis with key components Trina Solar, Pristine Sun LLC. MarketWatch 2019.
- [10] Carvallo J, Taneja J, Callaway D, Kammen DM. Distributed Resources Shift Paradigms on Power System Design, Planning, and Operation: An Application of the GAP Model. Proceedings of the IEEE 2019:1–17. https://doi.org/10.1109/JPROC.2019.2925759.
- [11] Frick NM, Schwartz LC, Taylor-Anyikire AM. A Framework for Integrated Analysis of Distributed Energy Resources: Guide for States. Berkeley, CA: Lawrence Berkeley National Laboratory; 2018.
- [12] Tamimi B, Cañizares C, Bhattacharya K. System Stability Impact of Large-Scale and Distributed Solar Photovoltaic Generation: The Case of Ontario, Canada - IEEE Journals & Magazine. IEEE Transactions on Sustainable Energy 2013;4. https://doi.org/10.1109/TSTE.2012.2235151.
- [13] Haghighat H, Kennedy SW. A Bilevel Approach to Operational Decision Making of a Distribution Company in Competitive Environments. Institute of Electrical and Electronics Engineers 2012;27.
- [14] Canova A, Giaccone L, Spertino F, Tartaglia M. Electrical Impact of Photovoltaic Plant in Distributed Network. 2007 IEEE Industry Applications Annual Meeting, 2007, p. 1450–5. https://doi.org/10.1109/07IAS.2007.223.
- [15] Kakimoto N, Piao Q-Y, Ito H. Voltage Control of Photovoltaic Generator in Combination With Series Reactor. IEEE Transactions on Sustainable Energy 2011;2:374–82. https://doi.org/10.1109/TSTE.2011.2148181.
- [16] Thomson M, Infield DG. Impact of widespread photovoltaics generation on distribution systems. IET Renewable Power Generation 2007;1:33–40. https://doi.org/10.1049/ietrpg:20060009.
- [17] Eftekharnejad S, Vittal V, Heydt GT, Keel B, Loehr J. Impact of increased penetration of photovoltaic generation on power systems. IEEE Transactions on Power Systems

2013;28:893-901. https://doi.org/10.1109/TPWRS.2012.2216294.

- [18] Palmintier B, Hale E, Hansen TM, Jones W, Biagioni D, Baker K, et al. Final Technical Report: Integrated Distribution-Transmission Analysis for Very High Penetration Solar PV. Golden, CO - USA: National Renewable Energy Laboratory (NREL); 2016. https://doi.org/10.2172/1239539.
- [19] Capitanescu F, Bilibin I, Romero Ramos E. A Comprehensive Centralized Approach for Voltage Constraints Management in Active Distribution Grid - IEEE Journals & Magazine. IEEE 2013;29. https://doi.org/10.1109/TPWRS.2013.2287897.
- [20] Ma C, Kaufmann P, Töbermann J-C, Braun M. Optimal generation dispatch of distributed generators considering fair contribution to grid voltage control. Renewable Energy 2016;87:946–53. https://doi.org/10.1016/j.renene.2015.07.083.
- [21] Saad NM, Sujod MZ, Ming LY, Abas MFB, Jadin MS, Ishak MR, et al. Impacts of Photovoltaic Distributed Generation Location and Size on Distribution Power System Network. International Journal of Power Electrics and Drive System 2018;9. https://doi.org/10.11591/ijpeds.v9n2.pp905-913 Nonlinear stochastic modeling for optimal dispatch of distributed energy resources in active distribution grids including reactive power.
- [22] Das CK, Bass O, Mahmoud TS, Kothapalli G, Mousavi N, Habibi D, et al. Optimal allocation of distributed energy storage systems to improve performance and power quality of distribution networks. Applied Energy 2019;252:113468. https://doi.org/10.1016/j.apenergy.2019.113468.
- [23] Mehrjerdi H, Hemmati R, Farrokhi E. Nonlinear stochastic modeling for optimal dispatch of distributed energy resources in active distribution grids including reactive power. Simulation Modelling Practice and Theory 2019;94:1–13. https://doi.org/10.1016/j.simpat.2019.01.005.
- [24] Georgilakis PS, Hatziargyriou ND. A review of power distribution planning in the modern power systems era: Models, methods and future research. Electric Power Systems Research 2015;121.
- [25] O'Shaughnessy E, Cutler D, Ardani K, Margolis R. Solar plus: Optimization of distributed solar PV through battery storage and dispatchable load in residential buildings. Applied Energy 2018;213:11–21. https://doi.org/10.1016/j.apenergy.2017.12.118.
- [26] Dominguez ODM, Kasmei MP, Lavorato M, Mantovani JRS. Optimal siting and sizing of renewable energy sources, storage devices, and reactive support devices to obtain a sustainable electrical distribution systems. Energy Systems 2018;9:529–50.
- [27] Linn C, Abbey C, Gil H, Kahrobaee S. Enhancing Distribution System Hosting Capacity through Active Network Management. 2018 IEEE Conference on Technologies for Sustainability (SusTech), Long Beach, CA, USA: IEEE; 2018, p. 1–6. https://doi.org/10.1109/SusTech.2018.8671376.
- [28] Haller M, Ludig S, Bauer N. Bridging the scales: A conceptual model for coordinated expansion of renewable power generation, transmission and storage. Renewable and Sustainable Energy Reviews 2012;16:2687–95. https://doi.org/10.1016/j.rser.2012.01.080.
- [29] Huang AQ, Crow ML, Heydt GT, Zheng JP, Dale SJ. The Future Renewable Electric Energy Delivery and Management (FREEDM) System: The Energy Internet. Proceedings of the IEEE 2011;99:133–48. https://doi.org/10.1109/JPROC.2010.2081330.
- [30] Karavas C-S, Kyriakarakos G, Arvanitis KG, Papadakis G. A multi-agent decentralized energy management system based on distributed intelligence for the design and control of autonomous polygeneration microgrids. Energy Conversion and Management 2015;103:166–79. https://doi.org/10.1016/j.enconman.2015.06.021.
- [31] Darghouth NR, Wiser RH, Barbose G, Mills AD. Net metering and market feedback loops: Exploring the impact of retail rate design on distributed PV deployment. Applied Energy 2016;162:713–22. https://doi.org/10.1016/j.apenergy.2015.10.120.
- [32] Erkoc M, Al-Ahmadi E, Celik N, Saad W. A game theoretic approach for load-shifting in the

smart grid. 2015 IEEE International Conference on Smart Grid Communications (SmartGridComm), Miami, FL, USA: IEEE; 2015, p. 187–92. https://doi.org/10.1109/SmartGridComm.2015.7436298.

- [33] Maharjan S, Zhu Q, Zhang Y, Gjessing S, Basar T. Dependable Demand Response Management in the Smart Grid: A Stackelberg Game Approach. IEEE Transactions on Smart Grid 2013;4:120–32. https://doi.org/10.1109/TSG.2012.2223766.
- [34] Wang J, Shahidehpour M, Li Z, Botterud A. Strategic Generation Capacity Expansion Planning With Incomplete Information. IEEE Transactions on Power Systems 2009;24:1002–10. https://doi.org/10.1109/TPWRS.2009.2017435.
- [35] Darghouth NR, Barbose G, Wiser R. The impact of rate design and net metering on the bill savings from distributed PV for residential customers in California. Energy Policy 2011;39:5243–53. https://doi.org/10.1016/j.enpol.2011.05.040.
- [36] Babacan O, Ratnam EL, Disfani VR, Kleissl J. Distributed energy storage system scheduling considering tariff structure, energy arbitrage and solar PV penetration. Applied Energy 2017;205:1384–93. https://doi.org/10.1016/j.apenergy.2017.08.025.
- [37] Laws ND, Epps BP, Peterson SO, Laser MS, Wanjiru GK. On the utility death spiral and the impact of utility rate structures on the adoption of residential solar photovoltaics and energy storage. Applied Energy 2017;185:627–41. https://doi.org/10.1016/j.apenergy.2016.10.123.
- [38] Carvallo JP, Shaw BJ, Avila NI, Kammen DM. Sustainable low-carbon expansion for the power sector of an emerging economy: the case of Kenya. Environmental Science & Technology 2017. https://doi.org/10.1021/acs.est.7b00345.
- [39] Fripp M. Switch: A Planning Tool for Power Systems with Large Shares of Intermittent Renewable Energy. Environ Sci Technol 2012;46:6371–8. https://doi.org/10.1021/es204645c.
- [40] He G, Avrin A-P, Nelson JH, Johnston J, Mileva A, Tian J, et al. SWITCH-China: A Systems Approach to Decarbonizing China's Power System. Environ Sci Technol 2016;50:5467–73. https://doi.org/10.1021/acs.est.6b01345.
- [41] Nelson J, Johnston J, Mileva A, Fripp M, Hoffman I, Petros-Good A, et al. High-resolution modeling of the western North American power system demonstrates low-cost and lowcarbon futures. Energy Policy 2012;43:436–47. https://doi.org/10.1016/j.enpol.2012.01.031.
- [42] Mai T, Barrows C, Lopez A, Hale E, Dyson M, Eurek K. Implications of Model Structure and Detail for Utility Planning: Scenario Case Studies Using the Resource Planning Model. Renewable Energy 2015:63.
- [43] Xu Q, Hobbs BF. Value of Model Enhancements: Quantifying the Benefit of Improved Transmis- sion Planning Models 2019:12.
- [44] Cole WJ, Marcy C, Krishnan VK, Margolis R. Utility-scale lithium-ion storage cost projections for use in capacity expansion models. 2016 North American Power Symposium (NAPS), 2016, p. 1–6. https://doi.org/10.1109/NAPS.2016.7747866.
- [45] OpenStreetMap contributors. Planet dump retrieved from https://planet.osm.org. 2019.
- [46] Schneider KP, Chen Y, Chassin DP, Pratt RG, Engel DW, Thompson SE. Modern Grid Initiative Distribution Taxonomy Final Report. 2008. https://doi.org/10.2172/1040684.
- [47] Hoke A, Butler R, Hambrick J, Kroposki B. Steady-State Analysis of Maximum Photovoltaic Penetration Levels on Typical Distribution Feeders. IEEE Transactions on Sustainable Energy 2013;4:350–7. https://doi.org/10.1109/TSTE.2012.2225115.
- [48] EIA. Electric power sales, revenue, and energy efficiency Form EIA-861 detailed data files. US Energy Information Administration 2016. https://www.eia.gov/electricity/data/eia861/ (accessed March 18, 2016).
- [49] 3TIER. Development of Regional Wind Resource and Wind Plant Output Datasets. Final Subcontract Report, 15 October 2007 - 15 March 2009. 2010. https://doi.org/10.2172/974458.

- [50] Perez R, Ineichen P, Moore K, Kmiecik M, Chain C, George R, et al. A new operational satelliteto-irradiance model. Sol Energy 2002;73.
- [51] Saha S, Moorthi S, Pan H-L, Wu X, Wang J, Nadiga S, et al. The NCEP Climate Forecast System Reanalysis. Bull Amer Meteor Soc 2010;91:1015–58. https://doi.org/10.1175/2010BAMS3001.1.
- [52] Barbose GL, Darghouth NR, LaCommare KH, Millstein D, Rand J. Tracking the Sun: Installed Price Trends for Distributed Photovoltaic Systems in the United States - 2018 Edition. 2018. https://doi.org/10.2172/1477384.
- [53] NREL. 2018 Annual Technology Baseline. Golden, CO: National Renewable Energy Laboratory; 2018.
- [54] EIA. Electric Power Monthly with data for July 2019. Washington DC, USA: Energy Information Administration; 2019.
- [55] Deryugina T, MacKay A, Reif J. The Long-Run Dynamics of Electricity Demand: Evidence from Municipal Aggregation. National Bureau of Economic Research; 2017. https://doi.org/10.3386/w23483.
- [56] Alberini A, Filippini M. Response of residential electricity demand to price: The effect of measurement error. Energy Economics 2011;33:889–95. https://doi.org/10.1016/j.eneco.2011.03.009.
- [57] Miller M, Alberini A. Sensitivity of price elasticity of demand to aggregation, unobserved heterogeneity, price trends, and price endogeneity: Evidence from U.S. Data. Energy Policy 2016;97:235–49. https://doi.org/10.1016/j.enpol.2016.07.031.
- [58] Sun Y, Yu Y. Revisiting the residential electricity demand in the United States: A dynamic partial adjustment modelling approach. The Social Science Journal 2017;54:295–304. https://doi.org/10.1016/j.soscij.2017.02.004.
- [59] Borenstein S. Electricity Pricing That Reflects Its Real-Time Cost 2009. /paper/Electricity-Pricing-That-Reflects-Its-Real-Time-Borenstein/5d1f1cb3585cd16b37b1aa55112b07f016f0eea6 (accessed March 21, 2020).